Dear editor

Thank you for giving us the opportunity to revise our article.

**Reviewer 1:**

The manuscript is innovative and elaborates on the risk of injury in endoscopic treatment of organs, but there are the following advice:

1. At the end of the article, a discussion should be added to mainly analyze the purpose of writing this article, whether there are methods to avoid or reduce organ injury. In operation, what prevention and treatment measures should be taken in clinical practice in the later stage, and what symptoms should be paid attention to.: A prevention section was added
2. And some other issues that need attention. Most of the references are from 5 to 10 years ago. It is recommended to update the literature from the last 5 years.: recent references were added and highlighted; reference number 1, 2, 34, 35

**Reviewer 2:**

This article provides a detailed description and discussion of abdominal organ injuries related to colonoscopy. The author lists potential injuries to the spleen, liver, pancreas, and other abdominal organs during colonoscopy and discusses their diagnostic and treatment methods. The article also mentions risk factors associated with the occurrence of injuries. Overall, the article offers important information about serious complications associated with colonoscopy and emphasizes the significance of close attention by clinical and endoscopic physicians to these complications. The content is informative, well-structured, and meets the requirements of academic writing. However, I noticed some areas for improvement:

1. Lack of introduction to colonoscopy: In the introduction, the author mentions that colonoscopy is a common diagnostic and therapeutic procedure but fails to provide a detailed introduction to the purpose, process, and applications of the procedure. To enhance readers’ understanding of colonoscopy, it is advisable to briefly introduce the basic principles and applications of this examination in the introduction. The introduction is improved and highlighted
2. Explanation of data sources and research methods: When describing the incidence of organ injuries related to colonoscopy, the author refers to some study results. However, the article does not provide specific information about these studies, such as study design, sample size, and research methods. To enhance the credibility of the article, it is recommended to provide more details about these studies. === Most of data were extracted from case reports because of the nature of the injuries, clinical trials cannot be done. Most of the relevant case reports were summarized in the tables.
3. Lack of discussion on preventive measures: While the article extensively describes the diagnostic and treatment methods for abdominal organ injuries related to colonoscopy, it fails to mention how to prevent these injuries. Given that colonoscopy is a routine procedure, preventive measures are crucial to reduce the occurrence of complications. It is suggested to include a discussion of methods and strategies for preventing these injuries in the discussion section.==== A prevention section was added

Reviewer 3:
I think this manuscript is OK. === thank you